Author's response to reviews

Title: A multilevel investigation of inequalities in clinical and psychosocial outcomes for women after breast cancer

Authors:

Philippa H Youl (pipyoul@cancerqld.org.au)
Peter D Baade (peterbaaade@cancerqld.org.au)
Joanne F Aitken (joanneaitken@cancerqld.org.au)
Suzanne K Chambers (suzanne.chambers@griffith.edu.au)
Gavin Turrell (g.turrell@qut.edu.au)
Christopher Pyke (c_pyke@mc.mater.org.au)
Jeffrey Dunn (jeffdunn@cancerqld.org.au)

Version: 2 Date: 12 September 2011

Author's response to reviews:

The additional files containing funding arrangements and ethical approval for the study are now attached.

As instructed a working version of the link for reference 51 has been provided and Prof and Assoc Prof have been deleted from the list of authors in the system.